Author’s response to reviews

Title: Activation of calpain-1 in human carotid artery atherosclerotic lesions

Authors:

Isabel Goncalves (isabel.goncalves@med.lu.se)
Mihaela Nitulescu (mihaela.nitulescu@med.lu.se)
Takaomi C Saido (saido@brain.riken.jp)
Nuno Dias (nuno.dias@med.lu.se)
Luis M Pedro (lmendespedro@clix.pt)
José Fernandes e Fernandes (jffernandes@netcabo.pt)
Mikko PS Ares (mikko.ares@pp.inet.fi)
Isabella Pörn-Ares (isabella.ares@helsinki.fi)

Version: 5 Date: 12 June 2009

Author's response to reviews: see over
June 12, 2009

Editorial Office
BMC Cardiovascular Disorders

Dear Editor,

Enclosed please find the latest version of our manuscript entitled “Activation of calpain-1 in human carotid artery atherosclerotic lesions”, which has been edited according to the suggestions of the editorial production team:

- The author list is written exactly as in the submission system
- Affiliations have been written in full
- We have indicated that all authors read and approved the final version
- The Email address list has been reformatted
- Tables have been redrawn
- Figure 1 has been cropped

I would be happy to assist if there are any further questions that need to be addressed before the publication of this manuscript.

Sincerely,

Isabella Pörn-Ares
PhD